Author’s response to reviews

Title: Lack of utility of risk score and gynecologic exam for screening for STIs in sexually active adolescents

Authors:

Eleuse MB Guimarães (eleusebrito@yahoo.com.br)
Mark DC Guimarães (drew@medicina.ufmg.br)
Maria Aparecida S Vieira (cidav@cultura.com.br)
Nádia M Bontempo (nadyamaciel@zipmail.com.br)
Mirian SS Seixas (seixasmirian@terra.com.br)
Mônica SD Garcia (monica_danda_garcia@yahoo.com.br)
Lyana ES Daud (lyanaelias@hotmail.com)
Rejane LM Côrtes (rejanemlc@uol.com.br)
Maria de Fátima C Alves (alves@iptsp.ufg.br)

Version: 2 Date: 20 February 2009

Author’s response to reviews:

Goiânia, February 20, 2009
Melissa Norton, MD
Editor-in-Chief, BMC Medicine

Dear,

Enclosed please find a manuscript entitled “Lack of utility of risk score and gynecologic exam for screening for STIs in sexually active adolescents”; by Eleuse Machado de Britto Guimarães, Mark Drew Crosland Guimarães, Maria Aparecida S. Vieira, Nádia M. Bontempo, Mirian S.S. Seixas, Mônica S.D. Garcia, Lyana. E. Santos, Rejane L.M. Côrtes and Maria de Fátima Costa Alves.

The manuscript was submitted to BMC Infectious Diseases and after revision was upgraded to BMC Medicine.

The authors undersigned are aware that the manuscript is being submitted to BMC Medicine and knowledge of and compliance with the Journal´s Editorials policies.

Since human subjects were involved in this study, we declare that the protocol was approved by the Ethics Committee of the University Hospital, Federal University of Goiás, Goiânia, Brazil.

Thank you for your attention.

Sincerely yours,
Eleuse Machado de Britto Guimarães, PhD Mark Drew Crosland Guimarães, PhD
Maria Aparecida S. Vieira, MD Nádia M Bontempo, MD
Mirian S. S. Seixas, MD Mônica S. D. Garcia, MD
Lyana Elias Santos Daud, MD Rejane L. M. Côrtes, MD
Maria de Fátima Costa Alves, PhD

Corresponding author:
Maria de Fátima Costa Alves
Universidade Federal de Goiás/Instituto de Patologia Tropical e Saúde Pública
Rua Delenda Resende de Melo S/N – 1o andar- Laboratório de Imunologia Celular
Setor Universitário CEP: 74605-050 Goiânia-GO, Brasil
e-mail: alves@iptsp.ufg.br